\newcommand{\noopsort}[1]{} \newcommand{\printfirst}[2]{#1}
  \newcommand{\singleletter}[1]{#1} \newcommand{\switchargs}[2]{#2#1}
\begin{thebibliography}{10}

\bibitem{attributor}
{Attributor - Subscription based web monitoring platform for content reuse
  detection}.
\newblock {http://www.attributor.com}.

\bibitem{berne}
{Berne Convention for the Protection of Literary and Artistic Works}.
\newblock {http://law-ref.org/BERN/index.html}.

\bibitem{n3logic}
Tim Berners-Lee, Dan Connolly, Lalana Kagal, Yosi Scharf, and Jim Hendler.
\newblock N3logic: A logical framework for the world wide web.
\newblock {\em Journal of Theory and Practice of Logic Programming}, 2007.

\bibitem{bliptv}
{blip.tv - Hosting, distribution and advertising platform for creators of web
  shows}.
\newblock {http://blip.tv/�}.

\bibitem{DBLP:conf/esws/BonattiDFNOPS06}
Piero~A. Bonatti, Claudiu Duma, Norbert~E. Fuchs, Wolfgang Nejdl, Daniel
  Olmedilla, Joachim Peer, and Nahid Shahmehri.
\newblock Semantic web policies - a discussion of requirements and research
  issues.
\newblock In {\em ESWC}, pages 712--724, 2006.

\bibitem{browser-market}
{Browser Market Share - 2009}.
\newblock {http://marketshare.hitslink.com/browser-market-share.aspx?qprid=0}.

\bibitem{by-legal-code}
{Creative Commons BY 3.0 Unported Legal Code}.
\newblock {http://creativecommons.org/licenses/by/3.0/legalcode}.

\bibitem{google_cc_search}
{Creative Commons Customized Search in Google}.
\newblock {Google Advanced Search Enables CC-Customized Searching,
  http://creativecommons.org/press-releases/entry/5692}.

\bibitem{liblicense}
{Creative Commons LibLicense library which integrates license metadata with
  content}.
\newblock {http://wiki.creativecommons.org/Liblicense}.

\bibitem{license-properites}
{Creative Commons License Properties}.
\newblock {License Properties,
  http://wiki.creativecommons.org/License-Properties}.

\bibitem{nc_user_study}
{Creative Commons Noncommercial study interim report,
  http://mirrors.creativecommons.org/nc-study/NC-Use-Study-Interim-Report-2009%
0501.pdf}.
\newblock {some title}.

\bibitem{copyright_debate}
{Debate on copyright and wrongs: "Existing copyright laws do more harm than
  good"}.
\newblock {http://www.economist.com/debate/days/view/310}.

\bibitem{license_babel}
{Different Open source Licenses}.
\newblock {http://www.opensource.org/licenses/alphabetical}.

\bibitem{exif}
{Exchangeable Image File Format }.
\newblock {http://www.exif.org/specifications.html}.

\bibitem{DBLP:conf/ccs/FeigenbaumFSS01}
Joan Feigenbaum, Michael~J. Freedman, Tomas Sander, and Adam Shostack.
\newblock Privacy engineering for digital rights management systems.
\newblock In Tomas Sander, editor, {\em Digital Rights Management Workshop},
  volume 2320 of {\em Lecture Notes in Computer Science}, pages 76--105.
  Springer, 2001.

\bibitem{xul}
{Firefox XUL Documentation}.
\newblock {https://developer.mozilla.org/En/XUL}.

\bibitem{flickrapi}
{Flickr API}.
\newblock {http://www.flickr.com/services/api}.

\bibitem{flickr_url}
{Flickr Farm URL}.
\newblock {http://www.flickr.com/services/api/misc.urls.html}.

\bibitem{flickroo}
{Flickr image reuse for openoffice.org}.
\newblock
  {http://wiki.creativecommons.org/Flickr-Image-Re-Use-for-OpenOffice.org}.

\bibitem{flickrlib}
{FlickrLib - Python wrapper for the Flickr API}.
\newblock {http://monotonous.org/2005/11/26/flickrlib-05/}.

\bibitem{hal08cc}
{Hal Abelson, Ben Adida, Mike Linksvayer, Nathan Yergler}.
\newblock {ccREL: The Creative Commons Rights Expression Language}.
\newblock {\em Creative Commons Wiki}, 2008.

\bibitem{DBLP:conf/policy/HansonBKSW07}
Chris Hanson, Tim Berners-Lee, Lalana Kagal, Gerald~J. Sussman, and Daniel~J.
  Weitzner.
\newblock Data-purpose algebra: Modeling data usage policies.
\newblock In {\em POLICY}, pages 173--177. IEEE Computer Society, 2007.

\bibitem{credit-on-flickr}
{How to attribute Flickr images}.
\newblock {http://www.squidoo.com/cc-flickr\#module12311035}.

\bibitem{ccval}
{Hugo Dworak, Creative Commons License Validation Service}.
\newblock {http://validator.creativecommons.org/}.

\bibitem{iptc}
{International Press Telecommunications Council Photo Metadata Format}.
\newblock {http://www.iptc.org/IPTC4XMP/}.

\bibitem{dist}
{Ivan Herman}.
\newblock {RDFa Distiller}, 2008.

\bibitem{harvey-thesis}
Harvey~C Jones.
\newblock Xhtml documents with inline, policy-aware provenance.
\newblock Master's thesis, Massachusetts Institute of Technology, May 2007.

\bibitem{jquery}
{JQuery Javascript Library by John Resig}.
\newblock {http://jquery.com}.

\bibitem{json}
{JSON - JavaScript Object Notation}.
\newblock {http://www.json.org/}.

\bibitem{air}
Lalana Kagal, Chris Hanson, and Daniel Weitzner.
\newblock Using dependency tracking to provide explanations for policy
  management.
\newblock {\em IEEE Policy 2008}, 2008.

\bibitem{attributor-prob}
{Ken Doctor, Blog Entry on "Attributor Fair Syndication Consortium Completes
  Newspaper Trifecta" }.
\newblock
  {http://www.contentbridges.com/2009/04/attributor-ad-push-on-piracy-complete%
s-newspaper-trifecta.html}.

\bibitem{qsign}
Jong~Wook Kim, K.~Sel�uk Candan, and Junichi Tatemura.
\newblock Efficient overlap and content reuse detection in blogs and online
  news articles.
\newblock In {\em 18th International World Wide Web Conference (WWW2009)},
  April 2009.

\bibitem{licensetagger}
{License Tagger}.
\newblock {http://wiki.creativecommons.org/Licensetagger}.

\bibitem{mozcc}
{MozCC - Firefox extension to discover Creative Commons licenses}.
\newblock {http://wiki.creativecommons.org/MozCC}.

\bibitem{newscredit}
{News Credit - WordPress plugin}.
\newblock {http://newscredit.org}.

\bibitem{operator}
{Operator Firefox Addon which discovers Microformats and RDFa in web pages}.
\newblock {https://addons.mozilla.org/en-US/firefox/addon/4106}.

\bibitem{owlmusicsearch}
{OWL Music Search}.
\newblock {http://www.owlmusicsearch.com/��}.

\bibitem{picscout}
{picScout - Image tracker for stock photography agencies and professional
  photographers}.
\newblock {http://www.picscout.com}.

\bibitem{powder}
{Protocol for Web Description Resources (POWDER)}.
\newblock {http://www.w3.org/2007/powder/}.

\bibitem{qdos}
{QDOS, Product developed by Garlik to measure the online status of an
  individual}.
\newblock {http://qdos.com/}.

\bibitem{rdf}
{RDF - Resource Description Framework}.
\newblock {http://www.w3.org/TR/rdf-syntax-grammar/}.

\bibitem{rdfa}
{RDFa - Resource Description Framework in Attributes}.
\newblock {http://www.w3.org/2006/07/SWD/RDFa/syntax/}.

\bibitem{scam}
N.~Shivakumar and H.~Garcia-Molina.
\newblock Scam: A copy detection mechanism for digital documents.
\newblock In {\em Second Annual Conference on the Theory and Practice of
  Digital Libraries}, 1995.

\bibitem{spinxpress}
{SpinXpress - Collaborative media production platform}.
\newblock {http://spinxpress.com}.

\bibitem{technorati}
{Technorati API}.
\newblock {http://technorati.com/developers/api/}.

\bibitem{thinkfree}
{Think Free - Java based web office suite}.
\newblock {http://www.thinkfree.com/}.

\bibitem{tab}
{Tim Berners-Lee}.
\newblock {Tabulator Redux: Browing and Writing Linked Data }.
\newblock In {\em {Linked Data on the Web Workshop at WWW08}}, 2008.

\bibitem{mozclip}
{Using the Mozilla Clipboard, Mozilla Developer Center}.
\newblock {https://developer.mozilla.org/en/Using-the-Clipboard}.

\bibitem{info-account}
Daniel~J. Weitzner, Harold Abelson, Tim Berners-Lee, Joan Feigenbaum, James
  Hendler, and Gerald~Jay Sussman.
\newblock Information accountability.
\newblock {\em Communications of the ACM}, June 2008.

\bibitem{danny_info_account}
Daniel~J. Weitzner, Harold Abelson, Tim Berners-Lee, Chris Hanson, James
  Hendler, Lalana Kagal, Deborah~L. McGuinness, Gerald~Jay Sussman, and
  K.~Krasnow Waterman.
\newblock Transparent accountable data mining: New strategies for privacy
  protection.
\newblock Technical Report MIT-CSAIL-TR-2006-007, Massachusetts Institute of
  Technology Computer Science and Artificial Intelligence Laboratory, January
  2006.

\bibitem{picscout-prob}
{William Faulkner, Tales From The IT Side: "PicScout, Getty Images and Goodbye
  iStockPhoto..!"}.
\newblock
  {http://williamfaulkner.co.uk/wordpress/2007/09/picscout-getty-images-and-go%
odbye-istockphoto/}.

\bibitem{xmp}
{XMP - Extensible Metadata Platform}.
\newblock {http://www.adobe.com/products/xmp/index.html}.

\bibitem{yahoo_cc_search}
{Yahoo Creative Commons Search}.
\newblock {http://search.yahoo.com/cc}.

\end{thebibliography}

\chapter{Background}
\label{background}

\section{Content Reuse Defined}

Content reuse, or in other words ``mash-ups" have existed for as long as content has existed. Musicians routinely use other songs and tunes in their compositions. Collage art is considered to be creative, and even original although it is composed from many different sources. Scientists routinely utilize data from different sources to conduct their own experiments. However, mash-ups, as we know them now, are a peculiarly digital phenomenon of the Web age. They are entirely a product made possible by the portable, mixable and immediate nature of digital technology. 

Reuse detection is important in domains such as plagiarism detection and even in biological sequence mining. Significant research has been carried out to detect reuse of text. This includes information retrieval techniques as mentioned in \cite{qsign, scam}, where the document is treated as a sequence of symbols and substring based fingerprints are extracted from the document to determine repetitive patterns. 

A potential legal problem arises when more than one legally encumbered content or data streams are bound together in the form of a mash-up. The users of the original content should remain within the bounds of the permitted use of the components comprising the mash-up. They can choose to ignore these permissions, or follow them. Either way, this creates a burden on them. Ignoring the license terms puts them at the peril of breaking the law, and following them slows the creative process.

\section{Policies for Rights Enforcement on the Web}

Policies governing the reuse of digital content on the Web can take several forms. It can be the \emph{Digital Rights Management} (DRM) approach exercised for example, by some commercial systems when distributing their media content. It could also be the \emph{Copyrights} alternative, that requires anybody reusing the content, either to have explicit permission from the original content creator, or express the original content creators rights as specified by the CC. 

\subsection{Digital Rights Management (DRM)}

Distribution and usage of copyrighted content is often controlled by up-front policy enforcement mechanisms such as DRM. These systems usually restrict access to the content, or prevent the content from being used within certain applications. The core concept in DRM is the use of digital licenses, which grant certain rights to the user. These rights are mainly usage rules which are defined by a range of criteria, such as frequency of access, expiration date, restriction to transfer to another playback device, etc. There are many applications that enforce DRM, such as Apple iTunes or Microsoft Windows Media Rights Manager. An example of a DRM enforcement would be a DRM software enabled playback device not playing a DRM controlled media transferred from another playback device, or not playing the media after the rental period for that media has ended. 

The use of DRM to express and enforce rights on content on the Web raises several concerns. First, the consumer privacy and anonymity are compromised. The authentication process in the DRM system usually requires the user to reveal her identity to access the protected content. This could lead to profiling of user preferences, and monitoring of user activity at large \cite{DBLP:conf/ccs/FeigenbaumFSS01}. 
%The other concern is the inability to use content under the �Fair Use� doctrine \cite{fair_use}. 
Another huge criticism of DRM is the usability of the content, where the user is limited to using proprietary applications to view or play the digital content requiring vendor lock-in.

\subsection{Copyright through Licensing}

Copyright is essentially the right to make copies. Prior to the Berne Convention, a multilateral treaty to which most countries now adhere \cite{berne}, content creators were required to explicitly give the \emph{copyright} (\copyright) notice to indicate that all rights are reserved for their works. Otherwise, their creations will fall under the public domain. However, Article 5(2) of the convention provides that ``the enjoyment and the exercise of [copyrights] shall not be subject to any formality" which means that any content that does not have any explicit license  will be protected by copyright law, whether the author is aware of it or not. Many argue this to be over inclusive to inhibit creativity \cite{copyright_debate}. For example, when reusing content that are copyrighted, the reuser faces many obstacles such as locating the copyright holder, obtaining permission, and usually having to pay royalty.

In the context of digital content, a `license' is describes the conditions of usage of copyrighted material. 
%Licenses are covered by federal copyright laws. 
A license on a digital file can exist whether or not there are any corresponding users for it. A user should abide by the license that covers the usage, and if any of the conditions of usage described in that license are violated, then the user should cease using that content. 

\subsubsection{Choosing a License}

Choosing a license can be very confusing. In the open source regime alone there are as many as 90 different licenses \cite{license_babel}. If choosing a license is tough for the creator, imagine what understanding the license is like for the end user.  

Creative Commons, a non-profit organization has been striving to provide a simple, uniform, and understandable set of licenses that content creators can use to issue their content under. These licenses provide a solution to the problem of copyright on the Web, while ensuring that the culture of reusing existing works to foster creativity is not hindered. Often, Web authors post their content with the understanding that it will be quoted, copied, and reused. Further, they may wish that their work only be used with  attribution, used only for non-commercial use, distributed with a similar license and will be allowed in other free culture media. To allow these use restrictions CC has composed four distinct license types: \emph{BY} (attribution), \emph{NC} (non-commercial), \emph{ND} (no-derivatives) and \emph{SA} (share-alike) that can be used in combinations that best reflects the content creator's rights.   

In order to generate the license XHTML easily, CC offers a license chooser that is hosted at \emph{http://creativecommons.org/license}. There, the users are given options to select for their works, and it will generate a snippet of XHTML that contains the RDFa \cite{rdfa} they need to embed when they are publishing the content on the Web.

\subsubsection{Creative Commons Rights Expression Language (ccREL)}

\emph{ccREL} \cite{hal08cc} is the standard recommended by the CC for machine readable expression of the meaning of a particular license. Content creators have the flexibility to express
their licensing requirements using this rights expression language and are not forced into choosing a pre-defined license for their works. Also, they are free to extend licenses to meet their own requirements. ccREL allows a publisher of a work to give additional permissions beyond those specified in the CC license with the use of the \emph{cc:morePermissions} property to reference commercial licensing brokers or any other license deed, and a \emph{dc:source} to reference parent works. Therefore, unlike in older CC recommendations, it is also possible to have content under a CC license that does not require attribution.

\subsubsection{Anatomy of a CC License in RDFa}

%Unclear - but correct
%\begin{figure}[!h]
%  \centerline{\epsfig{file=images/scenario_modified.jpg, width=1\linewidth}}
%  \caption{Illustration of Usage of CC Licensed Content}
%  \label{fig-cc-scenario-2}
%\end{figure}

%Clear Image
\begin{figure}[!h]
  \centerline{\epsfig{file=images/scenario.jpg, width=1\linewidth}}
  \caption{Illustration of Usage of CC Licensed Content}
  \label{fig-cc-scenario-2}
\end{figure}

Figure \ref{fig-cc-scenario-2} illustrates a simple CC license applied to an image. Suppose Alice is an avid Flickr user and she uploads her photos regularly. In her Flickr account settings she has applied \emph{CC-BY-2.0} to all her photos. This means she allows anybody to reuse her photos as long as they properly attribute her as given in her CC license terms.  Flickr does not yet markup this license information in RDFa, and is still under the older CC 2.0 specification. However, the CC license badge hyperlinked to the CC \emph{Deed Page} \footnote[1]{ A Deed Page is a human readable page that describes the license type.} of the corresponding license will be displayed in all of her photo album pages.

Supposing that Alice uses a service which marks up the license metadata in RDFa, the code snippet shown in Figure \ref{fig-cc-rdfa-license} shows how she would express her rights for one of her photos identified by the following URI:
http://flickr.com/photos/alice/somephoto.jpg. 
The RDF data are given in red.

\begin{figure}[!h]
\begin{alltt}
\singlespace{
<div about="\textcolor{red}{http://flickr.com/photos/alice/somephoto.jpg}">
<a rel="\textcolor{red}{license}" 
   href="\textcolor{red}{http://creativecommons.org/licenses/by-nc-nd/2.0/}"> 
   Creative Commons license 
</a>. 
If you use this photo within the terms of the license or make special 
arrangements to use the photo, please list the photo credit as 
<span property="\textcolor{red}{cc:attributionName}">\textcolor{red}{Alice}</span> 
and link the credit to <a rel="\textcolor{red}{cc:attributionURL}"
   href="\textcolor{red}{http://flickr.com/photos/alice}"> 
   http://flickr.com/photos/alice 
</a></div>}\end{alltt}
  \caption{License Expressed in RDFa}
  \label{fig-cc-rdfa-license}
\end{figure}

The important things to note in this code snippet is that Alice has given the subject (the URI of the photo) the following properties:
\begin{itemize}
\item \emph{attributionName}: `Alice'
\item \emph{attributionURL}: `http://flickr.com/photos/alice'.
\end{itemize}

If we assume that Bob uses Alice's photo in his blog or some other derivative work without embedding the XHTML that references Alice via the \emph{attributionURL} or the \emph{attributionName}, or does not at least mention the source of the photo - it is a clear violation of Alice's CC license terms.

\section{Data Purpose Algebra}

One of the hazards of combining multiple data sources is that incompatible licenses can get mixed up creating a license that basically freezes the creative process. Take for example a Non-Commercial (NC) license that gets mixed with a Share-Alike (SA) license. A SA license requires that the resulting product be shared under exactly the same conditions as the component product under SA. The resulting license in our scenario becomes NC-SA. But while the result satisfies the first license by also being NC, it fails the second license by not being only SA. So, if the component product under SA was unencumbered by the NC clause, adding it to an NC restricted component violates the SA requirement. Also we cannot simply ignore the NC clause and give the resulting work only the SA license. This is because somebody else might use the resulting derivative work which does not have the NC clause in some commercial use violating the rights of the original creator who composed the NC component. Many of these license combination conflicts are given in Figure \ref{fig-license-matrix}.

It has been shown that it is possible to model data usage policies programmatically by what is known as the \emph{Data Purpose Algebra} \cite{DBLP:conf/policy/HansonBKSW07}, by describing each content item $i$ in a data set, a source or agent that processes the data $Q_{\sc d}(i)$, the category of data $K_{\sc d}(i)$, and its purpose $P_{\sc d}(i)$. When another agent combines two or more data sets, a new data set is created whose content, category and purpose are some function of the agent, content, category and purpose of each of the component data sets. Specifically, if the agent or the source of the new data item is $a'$, the new category becomes the function ${\cal K}(K_{\sc d}(i))$ of the given category, and the allowed purposes of the new data item will be the more complex function ${\cal P}(P_{\sc d}(i), A_{\sc d}(i), a', K_{\sc d}(i))$ that may depend on the original purposes, the agents, and the category of the original data.

We believe that when reusing content on the Web,  the same principle could be applied. For the content item $i$ that is reused, the source function $Q_{\sc d}(i)$ would be represented by the URI of the content. The category of data $K_{\sc d}(i)$ will be represented by the content type (i.e. image, text, video, audio, etc.) that is being reused. The purpose $P_{\sc d}(i)$ will be determined by the CC license associated with it that specifies the allowed uses, restrictions and conditions. Then the function which composes the new set of purposes ${\cal P}(P_{\sc d}(i), A_{\sc d}(i), a', K_{\sc d}(i))$ should generate the XHTML that is required to embed the content with proper attribution.

\section{Inline Provenance using Metadata}

To be useful, metadata need to have three important characteristics: they have to be easy to produce, be embedded within the data they describe, and be easily readable. The easiest way to produce metadata is to have them be produced automatically. Any metadata that has to be produced manually by the user usually doesn't get produced at all. The easiest way to ensure that the link between metadata and the data they describe is not broken is by embedding the former inside the latter. This way, the two travel together inseparably  as a package. 
Finally, metadata have to be accessible easily, readable both manually as well as programmatically. At best, the metadata should be readable by crawlers of various search engines. Since metadata and data are traveling together, if popular search engines such as Google and Yahoo can read the metadata, by default the data become available to anyone who searches for it. RDF \cite{rdf} is the best known metadata format which satisfy all these criteria. It has lot of community support, adopted widely and is a W3C  recommendation.

Extensible Metadata Platform (XMP) \cite{xmp} is a technology that allows one to transfer metadata along with the content by embedding the metadata in machine readable RDF. This technology is widely deployed in embedding licenses in free-floating multimedia content such as images, audio and video on the Web.

Another format which is nearly universal when it comes to images is the Exchangeable Image File format (EXIF) \cite{exif}. International Press Telecommunications Council (IPTC) photo metadata standard \cite{iptc} is also another well known standard. The metadata tags defined in these standards cover a broad spectrum including date \& time information, camera settings, thumbnail for previews and more importantly, the description of the photos including the copyright information. However, these latter two formats do not store these metadata in RDF. They both use key-value pairs.

One major drawback of inline metadata formats such as XMP and EXIF is that, it is embedded in a binary file, completely opaque to nearly all users, whereas metadata expressed in RDFa will require colocation of metadata with human visible HTML. In addition to that, these metadata formats can only handle arbitrary number of properties using RDF predicates and these formats do not seem to take advantage of the rich expressivity offered by RDF.

\begin{landscape}
%\centering     % optional, probably makes it look better to have it centered on the page
\begin{figure}[!h]
  \centerline{\epsfig{file=images/license_matrix_new.jpg, height=4in}}
  \caption{License Composition Matrix for Detecting Share Alike Violations}
  \label{fig-license-matrix}
\end{figure}
The rows represent the resultant license of the composite work and the columns represent the license of a component content item. The \textcolor{green}{\ding{52}} symbol is used when the corresponding resultant license \textbf{\emph{can}} be given to the composite work if one of the components is under the license given by the column. However, if at least one of the cells is given the \textcolor{red}{\ding{56}} symbol, then the resultant license has a conflict with the corresponding component content item, and thus leads to a license violation. 
\end{landscape}

